Author's response to reviews

Title: The one-stop trial: Does electronic referral and booking by the general practitioner (GPs) to outpatient day case surgery reduce waiting time and costs? A randomized controlled trial protocol.

Authors:

Knut Magne Augestad (knut.magne.augestad@telemed.no)
Arthur Revhaug (arthur.reyhaug@unn.no)
Barthold Vonen (barthold.vonen@unn.no)
Roar Johnsen (roar.johnsen@medisin.ntnu)
Rolv-Ole Lindsetmo (RolfOle.Lindsetmo@UHhospitals.org)

Version: 2 Date: 23 July 2008

Author's response to reviews:

Norway 23 of July 2008

Dear BioMed Central Editorial Production Team

I have been abroad and have not been able to make the requested revision of my manuscript until today. I apologize for this.

I have now made the following changes of my manuscript

1) The list of authors in the manuscript are written exactly as they are in the submission system

2) Equal contributors are listed as requested

3) Additional files: The additional files were uploaded for proof of funding and ethical approval and were meant for the editorial team during the initial submission process.

These files are now deleted from the submission system.

4) Some small changes are made to text

Regarding the minor revisions I accept that the editorial production team make these changes on the authors behalf.
Thank you.

Yours sincerely

Knut Magne Augestad, MD

Corresponding author